

Tied migration and subsequent employment: Evidence from couples in Britain

Mark Taylor

ISER Working Paper 2006-05

Institute for Social and Economic Research

The Institute for Social and Economic Research (ISER) specialises in the production and analysis of longitudinal data. ISER incorporates the following centres:

- ESRC Research Centre on Micro-social Change. Established in 1989 to identify, explain, model and forecast social change in Britain at the individual and household level, the Centre specialises in research using longitudinal data.
- ESRC UK Longitudinal Studies Centre. A national resource centre for promoting longitudinal research and for the design, management and support of longitudinal surveys. It was established by the ESRC as independent centre in 1999. It has responsibility for the British Household Panel Survey (BHPS).
- European Centre for Analysis in the Social Sciences. ECASS is an interdisciplinary research centre which hosts major research programmes and helps researchers from the EU gain access to longitudinal data and cross-national datasets from all over Europe.

The British Household Panel Survey is one of the main instruments for measuring social change in Britain. The BHPS comprises a nationally representative sample of around 9,000 households and over 16,000 individuals who are reinterviewed each year. The questionnaire includes a constant core of items accompanied by a variable component in order to provide for the collection of initial conditions data and to allow for the subsequent inclusion of emerging research and policy concerns.

Among the main projects in ISER's research programme are: the labour market and the division of domestic responsibilities; changes in families and households; modelling households' labour force behaviour; wealth, well-being and socio-economic structure; resource distribution in the household; and modelling techniques and survey methodology.

BHPS data provide the academic community, policymakers and private sector with a unique national resource and allow for comparative research with similar studies in Europe, the United States and Canada.

BHPS data are available from the UK Data Archive at the University of Essex <u>http://www.data-archive.ac.uk</u>

Further information about the BHPS and other longitudinal surveys can be obtained by telephoning +44 (0) 1206 873543.

The support of both the Economic and Social Research Council (ESRC) and the University of Essex is gratefully acknowledged. The work reported in this paper is part of the scientific programme of the Institute for Social and Economic Research.

Acknowledgement:

Many thanks to John Ermisch, Birgitta Rabe, Stephen Jenkins, Steve Pudney and participants at seminars at ISER, ESPE 2005 and Policy Studies Institute for helpful discussions and comments on earlier drafts of this work. Financial support from the ESRC and the University of Essex is gratefully acknowledged. British Household Panel Survey data are available from the UK Data Archive (www.data-archive.ac.uk)

Readers wishing to cite this document are asked to use the following form of words:

Taylor, Mark (March 2006) 'Tied migration and subsequent employment: Evidence from couples in Britain', ISER Working Paper 2006-05. Colchester: University of Essex.

For an on-line version of this working paper and others in the series, please visit the Institute's website at: <u>http://www.iser.essex.ac.uk/pubs/workpaps/</u>

Institute for Social and Economic Research University of Essex Wivenhoe Park Colchester Essex CO4 3SQ UK Telephone: +44 (0) 1206 872957 Fax: +44 (0) 1206 873151 E-mail: <u>iser@essex.ac.uk</u> Website: <u>http://www.iser.essex.ac.uk</u>

© March 2006

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted, in any form, or by any means, mechanical, photocopying, recording or otherwise, without the prior permission of the Communications Manager, Institute for Social and Economic Research.

ABSTRACT

We use unique information on migration behaviour and the reasons for migration to study the impact of tied migration on labour market outcomes among husbands and wives. We find that fewer than 2% of couples migrate for job-related reasons, and that the majority of these move for reasons associated with the husband's job. Estimates from dynamic random effects models indicate that husbands and wives in couples that migrated for job-related reasons suffer lower job retention rates than non-migrants. Furthermore we find that tied migration reduces the probability of subsequent employment for both husbands and wives. In particular, tied migration has a large negative impact on job retention rates among wives.

NON-TECHNICAL SUMMARY

Empirical evidence demonstrates that family migration has large negative effects on labour market outcomes for women. In contrast, there is evidence of positive returns to migration among men. It is typically assumed that these differences are the result of family moves being undertaken to fulfil the career aspirations of the husband at the expense of those of the wife. In this paper we directly address issues relating to how family ties that force residential mobility affect labour market outcomes for men and women in couple households in Britain. Although there have been a number of British studies on how family migration affects an individual's career, our approach is unique in explicitly identifying tied migrants and distinguishing them from lead migrants. We identify tied and lead migrants using information on reasons for migration reported after any move, and in particular distinguish between migrants who moved for reasons associated with their own job or employment and those who moved for reasons associated with their spouse's job or employment. Our contribution to the literature is to (1) guantify the proportion of husbands and wives that are tied and lead migrants; (2) identify the impact of tied and lead migration on subsequent labour market status; and (3) examine whether these impacts differ between husbands and wives. Using panel data covering the period 1991-2003 from the British Household Panel Survey, we are able to take into account unobserved time-invariant individual-specific factors that are likely to be correlated with both the propensity to migrate and labour market status.

Our results indicate that job-related migration is not common. Fewer than 2% of couples in the sample migrated for reasons related to either the husband's or the wife's job or employment, representing one in four migrants. More than one half migrated for reasons relating to the husband's job or employment. Such couples moved an average of 107 kilometres, compared with 63 kilometres for those migrating for reasons to do with the wife's job and 30 kilometres for other, non-job related moves. Multivariate analysis indicates that job-related migration reduces job retention rates of both husbands and wives, but the size of this effect is larger for wives. Tied migration in particular reduces the probability of employment. Among husbands, this effect is the result of a combination of lower job entry and job retention rates relative to non-migrants while among wives it is the result of lower job retention rates relative to non-migrants. These results are largely robust to controlling for potential selection effects.

Therefore we find that tied migration has similar impacts on employment propensities irrespective of gender, although these result from different dynamics. This highlights the importance of identifying reasons for migration in assessing the impact of migration on labour market outcomes. Although employment-related migration among couples is uncommon, wives remain twice as likely as husbands to be tied migrants. The lower job retention rates among tied migrant wives results in a more widespread loss of occupational status and the associated pension rights. These have a longer term impact on the economic well-being of women. These analyses may understate the true impact of tied migration. Even tied migrants that remain in employment may suffer in terms of job quality, particularly if the move required a change in job or employer. In such circumstances tied migrants, although remaining employed, may suffer a relative wage loss, a decline in job satisfaction, or an increase in commuting time. We leave this for future research.

Tied migration and subsequent employment: Evidence from couples in Britain

Introduction

Empirical evidence demonstrates that family migration has large negative effects on labour market outcomes for women in terms of their labour market participation and employment, weeks worked, hours worked, income and wages, and attitudes towards work (Boyle et al 1999; Boyle et al 2001; Lee and Roseman 1999; Lichter 1980, 1983; Maxwell 1988; Morrison and Lichter 1988). The probability of unemployment or economic inactivity is higher for migrating married women than any other migrant group (Boyle et al 1999; Boyle et al 2002). In contrast, there is evidence of positive returns to migration among men (Bartel 1979; Yankow 2003; Böheim and Taylor 2005). It is typically assumed that differences between men and women in the impact of migration on labour market outcomes are the result of family moves being undertaken to fulfil the career aspirations of the husband at the expense of those of the wife. Not only do wives suffer in terms of their immediate career prospects, but the cumulative negative effect of repeated moves may contribute to the difference in economic and occupational status between men and women. Ultimately this difference explains much of the gender differential in pension rights and financial well-being in retirement.

In this paper we directly address issues relating to how family ties that force residential mobility affect labour market outcomes for men and women in couple households in Britain. Although there have been a number of British studies that address how family migration affects an individual's career, our approach is unique in explicitly identifying tied migrants and distinguishing them from lead migrants. We identify tied and lead migrants using information on reasons for migration reported after any move, and in particular distinguish between migrants who moved for reasons associated with their own job or employment and those who moved for reasons associated with their spouse's job or employment. Our contribution to the literature is to (1) quantify the proportion of husbands and wives that are tied and lead migrants; (2) identify the impact of tied and lead migration on subsequent labour market status; and (3) examine whether these impacts differ between husbands and wives.

Economic theory regards migration as an investment in human capital from which the individual receives a return in terms of their income or employment prospects. The labour market position of the migrant should be improved by the move. However in the context of the household, moves often occur to fulfil the career aspirations of the main breadwinner at the expense of those of their partner (Boyle et al 2001). Dual-earner couples have to make decisions regarding whether or not the careers of both partners should be pursued equally and if not then whose career should take precedence. Such decisions inevitably involve bargaining and compromise. Research has shown that married persons move long distances less frequently than single people, while dual-earner households migrate least (Böheim and Taylor 2002; Nivalainen 2004; van Ommerren et al 1999).

Human capital theory argues that relative earnings power and potential returns to migration determine which spouse is leading and which is tied (Mincer 1978).¹ Historically women have had lower occupational status and earnings than their male partners, and consequently household migration has typically been viewed as improving the career prospects of the husband with less regard to those of the wife. Mincer (1978) and Sandell (1977) argue that the effect of migration on married women is to reduce their post-migration labour force participation and the quality of their employment compared to their pre-migration levels. However, there is evidence suggesting that even in couples where women have a higher status occupation than their male partner migration does not benefit the female's career (Boyle et al 1999). A potential explanation for this is the perception of gender roles within the family, and in particular which of the husband and wife is perceived to play the role of principal earner (Bielby and Bielby 1992; Bird and Bird 1985). The issue of tied migrants and their labour market outcomes is becoming increasingly important in Britain given the rise in the proportion of dual-earner couples and multi-earner households (Gregg and Wadsworth 1996; Gregg at al 2004). As the labour market participation and career attachment of women continues to grow, it becomes more likely that wives will initiate family moves with the husband following.

¹ Hardill et al (1997) shows that in important, infrequent lifestyle decisions such as the location of residence, one partner normally tales the lead. Research on how intra-family bargaining affects an individual's welfare within families suggests that who receives income is an important determinant of who this is (Browning et al 1994; Fortin and Lacroix 1997; Chiappori, Fortin and Lacroix 2002).

However, empirical research is hampered by data limitations. Typically only actual migration and not mobility preferences or reasons for migration are observed, making it difficult to accurately distinguish tied migrants from lead migrants. Therefore it is typically assumed that the wife (husband) is the tied (lead) migrant, and as women are most likely to use their influence to prevent migration, husbands are assumed to be tied stayers (Smits 2001). We instead use data that not only allow identification of migrants but also their reasons for migration. We can therefore explicitly identify tied migrants as those who migrate for reasons associated with the job or employment of their spouse, and lead migrants as those who migrate for reasons associated with their own job or employment. We restrict our analysis to men and women living in couple households, and explicitly focus on couples who move together. Using panel data covering the period 1991-2003 from the British Household Panel Survey, we take into account unobserved time-invariant individual-specific factors that are likely to be correlated with both the propensity to migrate and labour market status. We also address issues relating to migrant self-selection.

Our results indicate that 7% of couple households moved house each year. However, fewer than 2% moved for job-related reasons, the majority of which were associated with the husband's job. Multivariate analyses indicate that impact of tied migration is similar for husbands and wives in that it reduces the probability of subsequent employment. However, more detailed analysis reveals that tied migrant husbands have both lower job entry rates and lower job retention rates than non-migrants, while tied migration reduces retention rates among wives.

Data

Panel data are required to accurately assess the impact of migration on labour market outcomes as they allow a comparison of the pre- and post-move situation. By tracing how an individual's labour market status changes following migration, we are able to identify which of the partners benefit and which, if any, suffer. Our analyses use the first 13 years of the British Household Panel Survey (BHPS), covering the period 1991-2003 (the latest year of data currently available). This is a nationally representative sample of some 5,500 private households recruited in 1991, containing approximately 10,000 adults. These same adults are interviewed each successive year. If anyone splits from their original household to form a new household, then all adult members of the new household are also interviewed. Children in original households are interviewed when they reach the age of 16. The core questionnaire

elicits information on income and earnings, labour market status, housing tenure and conditions, household composition, education and health at each annual interview. As part of maintaining the panel sample, information is collected on the migration behaviour of BHPS respondents, identifying those that move house and attempting to follow all migrants who remain in Britain. Although attrition rates among migrants are higher than among non-migrants, Buck (2000) reports that almost 75% of actual movers between 1991 and 1992 were traced compared to an overall response rate of 90%. Over the thirteen years of available panel data, an interview was possible with at least one household member in almost 80% of moving households.

We identify migrants from responses to the question: "Can I just check, have you yourself been living in this (house/flat) for more than a year?" In addition these data contain information about an individual's reasons for migration. At each date of interview individuals who have lived at their current address for less than a year are asked "Did you move for reasons that were wholly or partly to do with your own job or employment opportunities?" From responses to these questions we can identify migrants and job-related migrants. By matching responses of husbands and wives, we can identify couples that moved for reasons associated with the husband's job, those that moved for reasons associated with the wife's job and those that migrated for reasons associated with both jobs.² We define tied migrants as husbands in couples that migrated for reasons associated with the wife's job and wives in couples that migrated for reasons associated with the husband's job. Our null hypothesis is that tied movers suffer welfare losses while those that migrate for reasons associated with their own job or employment experience welfare gains and losses through labour market outcomes.

We focus explicitly on men and women in couples (both married and cohabiting), who we follow over time. In particular, we include in our sample couples in which both the husband and wife were aged 20 to 58 (inclusive), that were interviewed for at least two consecutive dates of interview and whose partnership remained intact with the same spouse at both dates

 $^{^{2}}$ In particular we identify couples that move for the husband's (wife's) job as those in which only the husband (wife) reported that the move was associated with their job.

of interview.³ We exclude those that were employed in the armed services at any time over the sample period, and full-time students. We use an unbalanced panel in the sense that couples enter and leave our sample as they enter and leave the relevant age range, or form and dissolve partnerships over time. Removing couples in which either partner has missing information on any variables used in the analysis results in a sample size of 5094 couples that contribute 25572 couple-year observations.

Table 1 shows that 7% of couples in our sample moved house each year.⁴ This table also indicates that 11% of migrating couples (0.7% of all couples) moved for reasons associated with only he husband's job, while 4% (0.3% of all couples) migrated for reasons associated with only the wife's job. In 5% of migrating couples (0.4% of all couples), both husband and wife reported that the move was associated with their employment.⁵ Therefore these data indicate about 16% of couples moved for reasons at least partly associated with the husband's job, considerably lower than the 28% of all working age migrant men experiencing job-related migration reported in Böheim and Taylor (2005). This is prima facie evidence that married men have lower job-related migration rates than single men and are therefore more likely to be tied stayers.

This table also indicates that 53% of migrating wives were tied migrants in the sense that the move was related to only the husband's job, compared with 21% of migrating husbands. Therefore despite increasing equality between men and women in career attachment and

³ We exclude couples where either partner were aged less than 20 because one or both may still be participating in education and any migration decision may not reflect expected labour market outcomes. We exclude those over the age of 58 because of the possibility that migration decisions may be based on retirement location preferences. Restricting the sample to couples that remained intact at two consecutive dates of interview may introduce some selection biases, with couples that disagree in their mobility preferences being more likely to separate. Ermisch (2003) finds that on average fewer than 3% of couples identified in the BHPS over the period 1991-2000 suffer a partnership break each year.

⁴ This includes all changes of address, irrespective of the distance moved. The proportion of couple households that move is lower than the 10% average for the working age population as a whole reported in Böheim and Taylor (2002). This indicates that either it is more difficult for married than single men and women to migrate or that couples initially locate in areas more likely meet their longer term requirements. We also retain multiple moves in our sample. This is for two reasons. Firstly, the first observed moved observed in the data (i.e. since 1991) is unlikely to be the first move in the respondent's life and so discarding second or further moves is to some extent an artificial distinction. Secondly focusing on first observed move would reduce already small sample sizes even further. However focusing on first observed moves does not change our substantive results.

⁵ Throughout we refer to the male partner in a couple as the husband and the female partner as the wife for simplicity, despite the fact that some couples were cohabiting rather than married. The BHPS data also contain information on migration preferences – identifying individuals who prefer to move from their current accommodation and those that prefer to stay. Analysis of these preferences shows that job-related migrants were less likely to report wanting to move than other migrants, which may indicate that job-related migration is largely unanticipated.

labour market experiences, wives are still twice more likely than husbands to be tied migrants. We also find that, on average, migrating couples moved a distance of 30 kilometres – indicating that many moves are of short-distance (see also Böheim and Taylor 2002).⁶ However, job-related migrants move significantly further distances (an average of 96 kilometres). Couples who moved for reasons related to the husband's job moved an average of 107 kilometres, while those who moved for reasons related to their wife's job moved an average of 63 kilometres. This indicates that job-related moves are of substantially longer distance than other (mostly accommodation-related) moves.

We are interested in whether tied migrants were less likely than lead migrants and other individuals to be in employment post-move, and whether this differs between men and women. To examine this, we construct a variable that indicates whether or not the respondent was in full-time or part-time work during the week prior to interview and relate this to the couple's migrant status. Table 2 summarises husbands and wives employment status at t-1 (prior to any move) by migrant status between t-1 and t. This indicates that among husbands job-related migrants were more likely to be in both full-time and part-time work at t-1 than non-migrants (89% were in full-time work, compared with 86% of non-migrants, and 5% were in part-time compared with 3% of non-migrants).⁷ Furthermore, a larger proportion of husbands that became lead migrants than became tied migrants were in full-time work, while tied migrants were more likely to be in part-time work.

A similar pattern emerges among wives. Wives in couples that migrated for job-related reasons were significantly more likely than non-migrants to be in full-time employment at t-1 (52% compared with 41%), but less likely to be in part-time employment (24% compared with 34%). Furthermore, we find that 72% of wives that became lead migrants were in full-time work at t-1, compared with 40% of those that became tied migrants. In contrast, 26% of those that became tied migrants were in part-time employment at t-1, compared with 21% of those that became lead migrants. This table therefore indicates that among both husbands and wives, full-time employment rates where higher among those that became lead migrants, and were lower among those that became tied migrants.

⁶ The distances are measured (in kilometres) between the full postcode of residence at date of interview at time t-1 and the full postcode of residence at date of interview at time t, rounded to the nearest 10 metres. Unfortunately these data are currently only available up to wave 12 of the data, and so we do not make further use of them in our analyses.

In Table 3 we summarise husbands and wives employment status at t (after any move) by migrant status between t-1 and t. This indicates that for husbands, full-time employment rates were highest for lead migrants (at 90%), and lowest for tied migrants (at 81%). Part-time employment rates were also highest for lead migrants (at 5%) and for husbands in couples that moved for both the husband's and wife's job (at 6%). A comparison with the results in Table 2 shows that among husbands full-time employment rates fell for all migrant groups, while part-time employment rates increased among lead migrants. For wives, lead migrants had the highest full-time employment rates at t (at 68%), while tied migrants had the lowest (33%). Part-time employment rates were highest for non-migrants (35%), and for tied migrants and wives in couples that migrated for reasons associated with both partners jobs (26%). They were lowest for lead migrants (21%). A comparison with Table 2 indicates that for wives full-time employment rates fell for job-related migrants (from 52% to 48%), and for both tied migrants (from 40% to 33%) and lead migrants (from 72% to 68%). Part-time employment rates did not change.

Table 4 focuses on employment inflow and outflow rates by migrant status.⁸ The inflow rate is defined as the probability of employment (either full- or part-time time) at *t* conditional on non-employment at t-1. The outflow rate is defined as the probability of non-employment at t conditional on employment at t-1. Among husbands, 22% of non-migrants and 24% of migrants entered employment between t-1 and t, while 3% of each left employment. Job-related migrants had significantly higher inflow rates into work and outflow rates from work than non-migrants. Husbands that were lead migrants had the highest employment inflow rate (at 71%), while tied migrants had a higher than average probability of leaving work (9.4%). Therefore husbands that were tied migrants were three times more likely to leave work as non-migrants, while lead migrants were significantly more likely to enter work than non-migrants.

Migrating wives had higher exit rates from employment than non-migrating wives -10% of migrants left work, compared with 6% of non-migrants. Job-related migrants had significantly higher employment inflow and outflow rates than non-migrants, at 31% and

⁷ However these differences are not statistically significant.

⁸ Unfortunately sample sizes are insufficient to allow detailed analysis of transitions by full- and part-time employment.

14%. Employment inflow rates were highest for wives in couples that migrated for reasons associated with both their own job and that of their husband (at 63%). Wives that were tied migrants had the highest outflow rates at 21%. Therefore wives who were tied migrants were four times more likely to leave work than non-migrants.

Table 5 presents migration rates between t-1 and t by the employment status of the couple at t-1. This indicates that couples in which neither the husband nor the wife was in employment had the greatest migration rates at 8%. Dual earner couples had the lowest migration rates at 6.3%. However, both single and dual earner couples were more likely to migrate for job-related reasons than jobless couples – about 1.3% of working couples migrated for job-related reasons compared with 0.4% of jobless couples. This may reflect labour market detachment among jobless couples. A larger proportion of single earner than dual earner couples migrated for reasons associated with the husband's job (1% compared with 0.6%).

These descriptive statistics indicate that (1) job-related migrants tend to move longer distances than other migrants; (2) a larger proportion of wives than husbands are tied migrants; (3) job-related migrants experience very different employment dynamics than other migrants; and (4) individuals who are tied movers in the sense that they move for their spouse's job suffer a labour market disadvantage relative to other migrant groups and non-migrants. In the remainder of the paper we examine the extent to which these patterns remain in a multivariate framework.

Estimation procedures and model specifications

Our modelling procedure focuses on the probability of an individual being in employment at a given date of interview, conditional on their labour market status one year ago and their migrant status. It is often argued that the returns to migration among married women may be underestimated because women who are more likely or more determined to succeed in the labour market are less likely to move for the benefit of their husband's career at a cost to their own. Through identifying both lead and tied migrants, we are able to examine this in detail. We use the panel nature of the data and panel data estimation techniques to allow for timeinvariant unobserved individual-specific effects (such as motivation, career attachment etc.) that are potentially correlated with both migrant status and the labour market outcomes of interest. We investigate the impact of the couple's migrant status on the probability of the husband and wife being employed. Our observed dependent variables, y_{ht} and y_{wt} , are binary, taking the value one if the husband (*h*)/wife (*w*) is in work (full-time, part-time or self-employment) at interview *t*, and zero otherwise.⁹ We estimate separate models for both husbands and wives, and for moving for job reasons generally and for reasons related to the husband's and wife's job in particular. The models are specified as

$$y_{ht}^* = x_{ht}'\beta_h + \alpha_h M_t + \gamma_h y_{ht-1} + u_{ht}$$
[1]

$$y_{wt}^* = x_{wt}' \beta_w + \alpha_w M_t + \gamma_w y_{wt-1} + u_{wt}$$
⁽²⁾

$$y_{ht}^* = x_{ht}'\beta_h + \alpha_{ht}TM_{ht} + \alpha_{hl}LM_{ht} + \alpha_{hb}MB_{ht} + \gamma_h y_{ht-1} + u_{ht}$$
[3]

$$y_{wt}^{*} = x_{wt}' \beta_{w} + \alpha_{wt} T M_{wt} + \alpha_{wl} L M_{wt} + \alpha_{wb} M B_{wt} + \gamma_{w} y_{wt-1} + u_{wt}$$
[4]

where y_t^* denotes the unobservable propensity of the husband/wife to be employed at time *t*, *x* is a vector of observable characteristics that influence y^* . In specifications [1] and [2] *M* indicates whether or not the couple moved for job-related reasons. In specifications [3] and [4] *TM* indicates whether or not the individual was the tied migrant, *LM* indicates whether or not the individual was the tead migrant and *MB* indicates whether or not the couple moved for reasons associated with the jobs of both partners (note that these three indicators are mutually exclusive).¹⁰ y_{t-1}^* is the employment status at the previous date of interview, β , α and γ are (vectors of) coefficients to be estimated and u_{it} is a random error.¹¹ An individual is observed to be in work when his/her propensity to be employed is greater than zero (i.e. $y_t^* > 0$). We model observed employment status at time t as a function of status in the previous period y_{t-1} to allow work experience at the previous interview to have a direct

⁹ We have also estimated different types of models, including multinomial logits (where the dependent variable takes the value 0 if the respondent is in work, 1 if economically inactive and 2 if unemployed), and tobit models of labour supply where the dependent variable is the number of hours worked (censored at 0 for non-workers). The results from these models are consistent with those presented here. We have also estimated models where the dependent variable takes the value 1 if the individual is in full-time work and zero otherwise. Again the results from doing so are qualitatively similar to those presented here.

¹⁰ Using this notation, the husband (wife) is a tied migrant if the couple moved for reasons associated with the wife's (husband's) job, and a lead migrant if they moved for reasons associated with the husband's (wife's) job.

¹¹ A valid criticism of this procedure is it assumes the employment decisions of husbands and wives are independent. We have also estimated (pooled) bivariate probit models to allow labour market participation decisions of husband and wife to be jointly determined through correlation between the error terms. Although confirming this interdependence in the labour market participation decisions of husband and wife, the results from such models are qualitatively similar to those presented here, and are available from the author on request.

effect on the probability of working at the current interview.¹² This is likely to be important in the current context as we might expect migration status to be strongly correlated with labour market attachment. Husbands and wives who exhibit strong (weak) attachment to the labour market and are (not) committed to their careers will have a high (low) probability of employment at t-1 and a low (high) probability of becoming a tied mover. Unless work experience at the previous date of interview is controlled for in the estimation procedure, the coefficients of main interest will be biased.

In our specifications we control for both observable and unobservable individual characteristics by decomposing the error terms u_t in the following way:

$$u_{ht} = \varepsilon_h + v_{ht}$$
$$u_{wt} = \varepsilon_w + v_{wt}$$

Where the ε denote the individual-specific unobservable effects and v are random error terms. We treat the ε as random and use the random effects probit model estimated under the common assumption that $v \sim IN(0, \sigma_v^2)$ and the v are independent of the other covariates.

This framework assumes that the time-invariant unobserved individual-specific effects (ε) are independent of the observable characteristics. This is quite unrealistic in our case as, for example, individuals who are less committed to their careers will be more likely to be a tied migrant, less likely to be a lead migrant, and less likely to be in employment. In this case the estimated coefficients of interest (α) will pick up some of the effects of the unobservable ε . To avoid this problem, we relax the assumption that the ε are independent of the observable time-varying characteristics. Following Chamberlain (1984), we model the dependence between ε and the observable characteristics by assuming that the regression functions of ε are linear in the means of all the time-varying covariates. Using equation [1] as an example, for husbands this can be written:

¹² Note that we do not attempt to provide a definitive interpretation of estimated coefficients on the lagged dependent variables. Positive coefficients may arise because (a) of genuine state dependence in employment, (b) of duration effects arising because the current employment spell is part of a single spell of work already in progress at the previous interview, (c) the observed characteristics of the individuals and the unobserved individual components do not adequately control for differences in individual employment propensities, or (d) of serial correlation in unobservables. As we are not concerned with the true causes of observed persistence in employment we do not attempt to disentangle these different effects.

$$\varepsilon_h = a_h + b_h' \overline{x}_h + c_h \overline{M} + \mu_h$$

An analogous formulation can be written for wives. We assume that the μ are independent of the x, M and u_t, a₀ is the intercept, $\overline{x_h}$ refers to the vector of means of the time varying covariates for husband h over time, and \overline{M} refers to the vector of means of the migration indicators for husband h over time. The coefficients in b_h and c_h corresponding to the timeinvariant variables are set equal to zero. Equations [1] and [2] therefore become:

$$y_{ht}^* = x_{ht}'\beta_h + \alpha_h M_t + \gamma_h y_{ht-1} + b_h' \overline{x}_h + c_h \overline{M} + \mu_h + v_{ht}$$
[5]

$$y_{wt}^{*} = x_{wt}' \beta_{w} + \alpha_{w} M_{t} + \gamma_{w} y_{wt-1} + b_{w}' \overline{x}_{w} + c_{w} \overline{M} + \mu_{w} + v_{wt}$$
[6]

h, *w*=1,...,*N*, *t* = 1,...,*T_i*, where we have absorbed the intercept into the β . This is equivalent to the random effects probit with additional regressors, \overline{x} and \overline{M} .¹³ Equations [3] and [4] are specified analogously. The correlation between two successive terms for the same individual is a constant, given by:

$$\rho = corr(u_{it}, u_{it-1}) = \frac{\sigma_{\mu}^2}{\sigma_{\mu}^2 + \sigma_{\nu}^2}$$

Model specifications

In our multivariate analyses we take into account lifecycle factors such as the birth of a child, a child's entry into primary or secondary school, or exit from the parental home, in modelling labour market outcomes. These are likely to have an impact both on migration decisions and on the labour market participation of particularly the wife. The decision to have a child and the ageing process of the child introduces complex issues and decisions involving the labour market participation of the mother and the level of satisfaction with the current place of residence. Other control variables are included to capture labour market attachment, job search intensity, job offer arrival rates and job retention rates. These include the employment status of the spouse at t-1, whether the couple are cohabiting (rather than married), their housing tenure, region of residence, age of the husband, highest educational qualification of the husband, ethnicity of the husband and the wife, the age difference between the husband

¹³ Our specifications treat the initial conditions as exogenous. This might be unrealistic – the labour market status of husbands and wives when first observed in the data may be the result of previous migration decisions (unfortunately we have no information on migration histories). We have also estimated more complex models that endogenise the initial condition (following Heckman 1981; Arulampalam et al 2000; Arulampalam 2002). These specify a reduced form equation for the initial condition using pre-sample information as identifying restrictions. The results from doing so are consistent with those presented herein.

and wife, whether the wife had higher educational qualifications than the husband (all measured at time t) and the lagged dependent variable.¹⁴

Results

The results from the estimated models are presented in Tables 4 and 5. In each case, the reference group of interest is husbands/wives that did not migrate (couples that migrated for non-employment reasons have been excluded). Note that we have estimated two specifications. The first includes the variables indicating couples migrant status to examine its impact on husbands and wives propensity to be employed. The second also includes these indicator variables interacted with labour market status at the date of interview prior to migration, to examine whether employment entry and retention rates differ by migrant status.

Table 5 presents the results from dynamic random effects probit models with job-related migration as the explanatory variable of interest.¹⁵ Focussing initially on the estimates for husbands, the first specification indicates that job-related migrants were less likely to subsequently be in employment than other husbands, all else equal. The coefficient is negative and statistically different from zero at conventional levels indicating that men in couples that migrated for job-related reasons were less likely than non-migrants to be subsequently employed. The second specification suggests that this negative relationship is caused primarily by lower employment stability among job-related migrants. The coefficient on the employed at t-1 and job-related migrant interaction term is negative and statistically significant with a point estimate of 0.202-0.625=-0.423 relative to a similar non-migrant. For wives, the estimates also indicate that job-related migrants were significantly less likely than non-migrants to be in employment at t. The negative coefficient is highly statistically significant. The second specification indicates that, as for husbands, this is caused by lower levels of job stability. Job-related migration increases the inflow rate into employment among wives (significant at the 11% level), but reduces job retention rates. The point estimate for job-related migrants in employment at t-1 can be calculated as 0.280-0.986=-0.706 relative to similar non-migrants. Furthermore, there is some evidence that job-related migration has a larger negative effect on job retention rates for wives than for husbands (a chi-squared test for

¹⁴ We have also estimated less parsimonious models that include job characteristics of the husband and wife (if employed) at t-1. The inclusion of such variables has little impact on the coefficients of interest, but reduces the size and significance of the coefficient on the lagged dependent variables.

equality of coefficients yields a p-value of 0.0635), although employment entry rates are no different.

The coefficients on other variables indicate strong persistence in employment among both men and women, particularly for those in full-time work - the coefficients on the lagged dependent variables are very large and highly statistically significant, and similar in magnitude for husbands and wives. There is also evidence of interdependence in labour supply decisions of husbands and wives. Men and women with employed spouses are significantly more likely to be in work themselves. Coefficients on the education variables are consistent with human capital theory and positive assortative mating. A monotonic relationship emerges between the husband's level of education and his probability of employment with the most highly educated having the largest probability of being in work. The husbands level of education also has positive and statistically significant impacts on the employment probability of the wife. Wives of highly educated men have a higher probability of employment than those married to less qualified men. Furthermore, the probability of employment among wives is increased if they have a higher level of education than their husband. Husbands of prime working age (35-44) and wives of men of prime working age are more likely to be in work than men (and wives married to men) aged below 35. Also, the probability of the husband being employed falls with each year older he is than his wife.

Family effects also emerge, especially among wives. The results indicate that having one child increases the probability of employment among husbands relative to having no children, but has no impact on the probability of employment among wives. However, it is the timing and age structure of the family that has the largest impacts on the probability of employment among wives. In particular, having a pre-school age child reduces the probability of employment, and mothers of recently born first children have a much reduced probability of current employment than non-mothers and mothers of older children.

Table 6 shows the results from distinguishing between the impacts of tied and lead migration on the employment propensities of husbands and wives. Specification [1] indicates that husbands that were tied migrants were significantly less likely to be employed than non-

¹⁵ Note that the estimated values for rho are 0.10 for men and women. This indicates that 10% of the variance in the dependent variable is explained by the individual-specific unobserved effect.

migrants – the estimated coefficient is negative and well-determined. Furthermore, chisquared tests reveal that the estimated impact on the husband's employment status of being a tied migrant is significantly different from that of being a lead migrant (a p-value of 0.0561). Therefore husbands who were tied migrants suffer in terms of their own employment both relative to non-migrants and relative to those who were lead migrants.

Specification [2], which includes the interaction terms, suggests that husbands that were lead migrants had higher entry rates into employment than non-migrants, and similar job retention rates. However although the sizes of the coefficients are relatively large, they are at best on the margins of statistical significance. The coefficient on being a tied migrant is also large and negative (although not well determined) suggesting men in couples that moved to follow their wife's career suffered in terms of lower entry rates to employment. Chi-squared test statistics indicate that the coefficients on being a tied migrant and on being a lead migrant on job entry rates are significantly different (p-value=0.0720). From this we conclude that tied migrant is harmful to a husband's labour market prospects relative to being the lead migrant. This highlights the importance of distinguishing the reasons behind migration in assessing its impact on subsequent labour market outcomes.

The estimates indicate that wives who were tied migrants suffered in terms of their subsequent employment – the coefficient is negative (-0.590) and well-determined (specification [1]). As for husbands, this indicates that tied migrant wives are less likely to be subsequently employed than non-migrants. Furthermore, the size of the effect is similar to that for husbands. Tied migration has a similar sized effect on the probability of employment for husbands and wives (a chi-squared test for the equality of α_{ht} and α_{wt} yields a p-value of 0.2686). It is not, however, significantly different from the impact on wives of being a lead migrant (a test that $\alpha_{wl}=\alpha_{wt}$ yields a p-value of 0.7381). The coefficient on the lead migrant indicator is also negative and statistically significant at the 10% level. Therefore tied migration and lead migration have similar impacts on subsequent employment among wives. Chi-squared tests of equality of coefficients across equations indicate that the impact on subsequent employment of lead migration among husbands differs from that of tied migration for wives (a test that $\alpha_{hl}=\alpha_{wt}$ yields a p-value of 0.0002).

The second specification for wives indicates that the negative impact on employment propensities of tied migration is due to significantly lower job retention rates – wives that were tied migrants were more likely than non-migrants to leave work. The coefficient on the tied migrant and employed at t-1 interaction term is negative and statistically significant. The point estimate can be calculated as 0.053-0.919=-0.866 relative to non-migrants. However, chi-squared tests indicate that we cannot reject the hypothesis that for wives, the coefficient on the tied migrant and employed at t-1 interaction term (-0.919) is equal to that on the lead migrant and employed at t-1 interaction term (-0.836), yielding a p-value of 0.9034. Therefore although wives that are tied migrants suffer in the labour market, we reject the hypothesis that they suffer more than those who are lead migrants. The results also indicate that wives in couples that moved to benefit both their own and their husbands' jobs had higher entry rates into employment than non-migrants (a coefficient of 1.207), and similar job retention rates (a point estimate of 1.207-1.598=-0.391).

Chi-squared tests comparing coefficients across equations indicate that the impact of lead migration on the probability of entering employment for husbands is significantly different from that of tied migration among wives – a test that $\alpha_{hl} = \alpha_{wt}$ yields a p-value of 0.0001. Furthermore, we find that tied migration has significantly different impacts on both employment inflow rates and job retention rates for husbands and wives. Chi-squared tests that $\alpha_{ht} = \alpha_{wt}$ yield p-values of 0.0047 and 0.0060. Tied migration reduces both the job entry and the job retention rates among husbands (although not significantly), but has a large, negative impact on job retention rates among wives (with little effect on entry rates).

The key results are summarised in Table 8. Job-related migration and tied migration in particular reduces employment stability rates among both husbands and wives. Differences emerge between husbands and wives in the impact of tied migration on job entry rates and job retention rates.

Addressing endogeneity and selection issues

When rational individuals decide whether or not to migrate, they choose the option that yields the greatest expected net utility gain. Positive selection into lead migration occurs if individuals who are more committed to their careers and who are more likely to be in employment are also more likely to move for reasons associated with their own job or

employment. Similarly, negative selection into tied migration occurs if individuals who are less committed to their careers and who are less likely to be in work are also more likely to move for reasons associated with their spouse's job or employment. Non-migrants expect to maximise their utility levels from staying at the same address. Our dynamic panel data model specifications which allow for time-invariant unobserved heterogeneity (correlated with observable characteristics) and the individual's labour market status prior to any migration minimise these selection effects. However, we now directly address some issues relating to endogeneity and self-selection in migrant status.

Firstly, we examine the labour market attachment prior to migration of husbands and wives in each migrant status. To do this, we focus on responses to the question:

'Thinking about the hours you work, assuming you would be paid the same amount per hour, would you prefer to work fewer hours, more hours, or continue to work the same number of hours?'

This question is asked of men and women in employment at each date of interview, and so these analyses necessarily focus on those in work at t-1. If individuals more attached to the labour market and to their career positively select into lead migration, we anticipate that a larger proportion of lead migrants will want to work more hours and a smaller proportion will want to reduce their work hours. If individuals who are less attached to the labour market and to their careers negatively select into tied migration, we anticipate those that become tied migrants will be more likely to want to reduce their working hours and less likely to want to increase their work hours. Table 9 displays the results from this exercise.

This table indicates that 40% of husbands wanted to reduce their work hours, 6% wanted to increase them and 54% wanted to continue to work the same hours. Husbands who were subsequently job-related migrants were more likely than average to want to work more hours -8.6% reported wanting to do so (although this difference is not statistically significant). So there is some evidence that among husbands, job-related migrants were positively selected. However, the subsequent columns indicate that it was tied migrants who were most likely to report wanting to increase their hours of work (13% did so) while lead migrants were more likely than average to report wanting to reduce their hours of work (46% did so – again these differences are not statistically significant at conventional levels). We also find that husbands

that become lead migrants worked on average 3.5 hours more per week than those that became tied migrants, which is some evidence of self-selection into migrant status. However, evidence suggests that this difference in working hours is caused by hours constraints in the job rather than a lack of labour market attachment.

There is more convincing evidence of self-selection among wives. In particular, we find that wives that become tied migrants were less likely than average to report wanting to work more hours (5.7% compared with 7.1%), while those that became lead migrants were more likely than average to report wanting to work more hours (8.8% compared with 7.1% – again these differences were not statistically significant). Furthermore, we find that wives that became lead migrants were working more hours per week on average than those that became tied migrants (35 hours per week compared with 30 hours).

To examine the impact of self-selection on our results, we have estimated two-stage selection correction models that explicitly control for non-random selection into each migrant group (see, for example, Lee and Roseman 1999). The first stage in these models is to estimate the propensity of couples to be in each of the migrant groups. The inverse Mill's ratios calculated from these migration equations are then used as additional regressors in the employment models to control for non-random selection. Identification of the variables in the employment models rely on including at least one variable in the selection models but not in the second stage employment regressions. In other words, it depends on having variables that determine the probability of couples migrating to follow the husband's or the wife's job but not the probability of employment conditional on migration. We use information collected at each date of interview on individuals migration preferences. In particular, each individual in the sample is asked at each date of interview whether or not they would prefer to move house or to stay in their current accommodation. If they would prefer to move, they are then asked the main reason for wanting to move. From these questions at the date of interview prior to any migration we identify in which couples the husband wanted to move for job reasons, the wife wanted to move for job reasons, and in which both husband and wife wanted to move for job reasons. We use these indicators as identifying variables in the migration models.¹⁶ We estimate separate models for couples migrating for job reasons, for reasons relating to the

¹⁶ These variables prove to be both individually and jointly insignificant when included in employment equations for the husband (p-value of 0.8387) and the wife (p-value of 0.8050).

husband's job, for reasons relating to the wife's job, and for reasons relating to both jobs and use these to generate the inverse Mill's ratios that are then used as regressors in the employment models.

The results from the migration equations are shown in Table 10. These indicate that the identifying variables are individually and jointly highly significant in each equation – wanting to move for job-related reasons at t-1 is a strong predictor of subsequent migration behaviour. Other significant determinants of migration status include the employment status of the wife (couples in which the wife was employed were more likely to move for reasons related to the wife's employment), the husband's education level (migration rates increase with education, and the probability of moving for reasons associated with the wife's job are increased if the wife is more highly educated than her husband), being a private tenant (which increases the migration propensity), and having pre-school age children in the household (which increases the propensity to move for job-related reasons and for reasons associated with the husbands job in particular). These are common findings in the literature (Böheim and Taylor 2002; Champion et al 1998; Clark and Dieleman 1996).

The results from the selectivity-corrected employment equations are shown in Table 11. The first panel indicates that controlling for potential selection effects has little impact on the estimates for husbands. For wives, the results suggests that there is some negative selection into job-related migration – the negative and statistically significant impact present in specification [1] of Table 6 disappears when allowing for selection effects. However, the negative and statistically coefficient on the job-related migrant and employed at t-1 term remains, indicating that wives that migrate for job-related reasons have lower levels of employment stability than otherwise similar non-migrants, even when controlling for potential selection effects.

Some evidence of selection effects also emerge in the second panel of Table 11. In particular, the negative impacts of being a tied mover apparent among both husbands and wives in the original models disappear when allowing for selection effects – husbands and wives were negatively selected into tied migration. However, the negative and statistically significant impact of tied migration on job retention rates among wives remains when allowing for selection effects – employed wives who were tied migrants are less likely than otherwise

similar non-migrants to remain in employment. These results are summarised in Table 12. The key results from the previous models remain – husbands and wives who migrate for job-related reasons, and tied migrant wives in particular – are significantly less likely to be subsequently employed than otherwise similar non-migrants.

Conclusions

In this paper we have investigated the labour market consequences of migration for men and women in couple households in Britain. Our contribution to this literature is to use unique information on reasons for moving to explicitly identify tied and lead migrants and to investigate the impact of such migration on the employment status of husbands and wives. BHPS data indicate that job-related migration was not common. Fewer than 2% of couples migrated for reasons related to either the husband's or the wife's job or employment, representing one in four migrants. More than one half of these migrated for reasons relating to the husband's job or employment. Such couples moved an average of 107 kilometres, compared with 63 kilometres for those migrating for reasons to do with the wife's job and 30 kilometres for other, non-job related moves.

Our analysis indicates that job-related migration reduces job retention rates of both husbands and wives, but the size of this effect is larger for wives. Tied migration significantly reduces the probability of employment for both husbands and wives, and the sizes of these effects are similar. Among husbands, this effect is the result of a combination of lower job entry and job retention rates relative to non-migrants while among wives it is the result of lower job retention rates relative to non-migrants. These results are largely robust to controlling for potential selection effects.

Therefore we find that tied migration has similar impacts on employment propensities irrespective of gender, although these result from different dynamics. This highlights the importance of the reasons for migration in assessing the impact of migration on labour market outcomes. Although employment-related migration among couples is uncommon, wives remain twice as likely as husbands to be tied migrants. The lower job retention rates among tied migrant wives results in a more widespread loss of occupational status and the associated pension rights. These have a longer term impact on the economic well-being of women. These analyses may understate the true impact of tied migration. Even tied migrants that

remain in employment may suffer in terms of job quality, particularly if the move required a change in job or employer. In such circumstances tied migrants, although remaining employed, may suffer a relative wage loss, a decline in job satisfaction, or an increase in commuting time. We leave this for future research.

References

- Arulampalam, W (2002) "State dependence in unemployment incidence: evidence for British men revisited", IZA Discussion Paper No. 630.
- Arulampalam, W., A.L. Booth and M.P. Taylor (2000), "Unemployment persistence", *Oxford Economic Papers*, 52, pp.24-50.
- Bartel, A.P. (1979), "The migration decision: What role does job mobility play?" American *Economic Review*, 69, 775-786.
- Bielby, W. T. and D. D. Bielby (1992). "I Will Follow Him: Family Ties, Gender-Role Beliefs, and Reluctance to Relocate for a Better Job." *American Journal of Sociology* 97(5): 1241-1267.
- Bird, G. A. and G. W. Bird (1985). "Determinants of Mobility in Two-earner Families: Does the Wife's Income Count?" *Journal of Marriage and the Family* 47(3): 753-758.
- Böheim, R. and M.P. Taylor (2002), "Tied down or room to move? Investigating the relationships between housing tenure, employment status and residential mobility in Britain", *Scottish Journal of Political Economy*, 49 (4), pp.369-392.
- Böheim, R. and M.P. Taylor (2005), "From the dark end of the street to the bright side of the road? The wage returns to migration in Britain", *Labour Economics* (forthcoming).
- Boyle, P., T.J. Cooke, K. Halfacree and D. Smith (1999), "Gender inequality in employment status following family migration in GB and the US: The effect of relative occupational status", *International Journal of Sociology and Social Policy*, 19, pp.115-150.
- Boyle, P., T.J. Cooke, K. Halfacree and D. Smith (2001), "A cross-national comparison of the impact of family migration on women's employment status", *Demography*, 38 (2), pp.201-213.
- Boyle, P., T.J. Cooke, K. Halfacree and D. Smith (2002), "A cross-national study of the effects of family migration on women's labour market status: Some difficulties with integrating microdata from two censuses", *Journal of the Royal Statistical Society Series A*, 165 (3), pp.465-480.
- Browning, M., F. Bourguignon, P-A Chiappori and V. Lachene (1994), "Income and outcomes: A structural model of intrahousehold allocation", *Journal of Political Economy*, 102, pp.1067-1097.
- Buck, N.H. (2000), "Using panel surveys to study migration and residential mobility", in D. Rose (ed), *Researching Social and Economic Change*, London: Routledge.
- Champion, A., P. Rees, P. Boyle and J.Stillwell (1998), The determinants of migration flows in England: a review of the existing data and evidence. A report prepared for the Department of the Environment, Transport and the Regions. Department of Geography, University of Leeds.
- Chiappori, P-A, B. Fortin and G. Lacroix (2002), "Household labor supply, sharing rule and the marriage market", *Journal of Political Economy*, 110 (1), pp37-72.
- Clark, W.A.V. and F.M. Dieleman (1996), *Households and housing: choice and outcomes in the housing market*. Centre for Urban Policy Research, The State University of New Jersey.
- Ermisch, J.F. (2003), "Family bargaining and residential mobility", mimeo, University of Essex.
- Fortin, B. and G. Lacroix (1997), "A test of the unitary and collective models of household labour supply", *Economic Journal*, 107, pp.933-955.
- Gregg, P., R. Scutella and J. Wadsworth (2004), "Reconciling workless measures at the individual and household level: Theory and evidence from the United States, Britain, Germany, Spain and Australia", CEP Discussion Paper 635.

- Gregg, P. and Wadsworth, J. (1996), "More work in fewer households?", In J. Hills (Ed.), *Income and Wealth: New inequalities*, Cambridge University Press, Cambridge.
- Ham, J.C., Li, X. and Reagan, P.B. (2003), Propensity score matching, a distance-based measure of migration, and the wage growth of young men. Mimeo, Ohio State University.
- Hardill, I., A.E. Green, A.C. Duddleston and D.W. Owen (1997), "Who decides what? Decision making in dual career households", *Work, Employment and Society*, 11 (2), pp.313-326.
- Heckman, J.J. (1981), "The incidental parameter problem and the problem of initial conditions in estimating a discrete time-discrete data stochastic process", in C. Manski and D. McFadden (eds), *Structural Analysis of Discrete Data with Econometric Applications*, MIT Press, Cambridge MA, pp.114-178.
- Lee, S. and C. C. Roseman (1999). "Migration Determinants and Employment Consequences of White and Black Families, 1985-1990." *Economic Geography* 75(2): 109-133.
- Lichter, D.T. (1980), "Household migration and the labor market position of married women", *Social Science Research*, 9, pp.83-97.
- Lichter, D.T. (1983), "Socioeconomic returns to migration among married women", *Social Forces*, 62, pp.489-503.
- Maxwell, N.L. (1988), "Economic returns to migration: Marital status and gender differences", *Social Science Quarterly*, 69, 108-121.
- Mincer, J. (1978), "Family migration decisions", *Journal of Political Economy*, 86, pp.749-773.
- Morrison, D. R. and D. T. Lichter (1988). "Family Migration and Female Employment: The Problem of Underemployment among Migrant Married Women." *Journal of Marriage and the Family* 50: 161-172.
- Nivalainen, S. (2004), "Determinants of family migration: Short moves vs long moves", *Journal of Population Economics*, 17, pp.157-175.
- Sandell, S.H. (1977), "Women and the economics of family migration", *Review of Economics and Statistics*, 59, pp.406-414.
- Smits, J. (2001), "Career migration, self-selection and the earnings of married men and women in the Netherlands, 1981-1993", *Urban Studies*, 38 (3), pp.541-562.
- Van Ommeren, J.N., P. Rietveld and P. Nijkamp (1999), "Impacts of employed spouses on job-moving behaviour", *International Regional Science Review*, 22 (1), pp.54-68.
- Yankow, J.J. (2003), "Migration, job change and wage growth: A new perspective on the pecuniary return to geographic mobility", *Journal of Regional Science*, 43 (3), 483-516.

Tuble 1. Moving status (1 to the Couple households							
Status	Ν	%	% movers	% job	Distance		
				movers	moved (kms)*		
Non-mover	23893	93.4					
Moved	1679	6.6	100.0		29.9		
Moved for job reasons	321	1.3	19.1	100.0	95.7		
Moved for husband's job	166	0.7	9.9	71.7	107.3		
Moved for wife's job	72	0.3	4.3	22.4	63.1		
Moved for both jobs	<i>83</i>	0.3	4.9	25.9	98.2		
Total	25572	100.0					

Table 1: Moving status t-1 to t: Couple households

Notes: BHPS 1991-2003. Numbers are person-years. *Distance moved is kilometres between the full postcode of residence at wave t-1 and the full postcode of residence at wave t, rounded to the nearest 10 metres. This information is not available for the wave 13 data, and so focuses only on migrants in the first 12 years of data.

Table 2: Employment rates at t-1 by migrant status t-1 to t: Men and women in couple households

	Hush	oands	Wives			
	Full-time	Part-time	Full-time	Part-time		
Non-migrants	85.7	3.4	41.0	34.2		
Migrants	85.2	3.2	45.8^{*}	25.8^{*}		
Job-related migrants	88.8	4.7	52.0^{*}	24.0^{*}		
Lead migrant	92.2^{*}	3.6	$72.2^{*\dagger}$	20.8^*		
Tied migrant	83.3	5.6	40.4^{\dagger}	25.9^*		
Moved for both jobs	86.8	6.0	57.8^{*}	22.9^{*}		
Total	85.7	3.4	41.3	33.6		

Notes: Employment rates. BHPS 1991-2003. * indicates employment rate significantly different from that for non-migrants at the 5% level. † indicates employment rate significantly different from that for job-related migrants at the 5% level.

Table 3: Employment rates at t by migrant status t-1 to t: Men and
women in couple households

	Hust	oands	Wi	ves
	Full-time	Part-time	Full-time	Part-time
Non-migrants	84.6	3.3	40.3	34.7
Migrants	83.9	3.7	42.9^*	26.9^{*}
Job-related migrants	86.6	4.4	47.7^{*}	24.6^{*}
Lead migrant	90.4^{*}	4.8	$68.1^{*\dagger}$	20.8^{*}
Tied migrant	80.6	1.4	33.1^{\dagger}	25.9^*
Moved for both jobs	84.3	6.0	59.0^{*}	25.3
Total	84.6	3.3	40.3	34.2

Notes: Employment rates. BHPS 1991-2003. * indicates employment rate significantly different from that for non-migrants at the 5% level. † indicates employment rate significantly different from that for job-related migrants at the 5% level.

nousenoius						
	Hus	bands	W	ives		
	Inflow rate	Outflow rate	Inflow rate	Outflow rate		
Non-migrants	21.8	2.9	17.8	5.8		
Migrants	24.2	3.4^{*}	20.5	10.1^{*}		
Job-related migrants	47.6^{*}	4.7^{*}	31.2^{*}	13.9*		
Lead migrant	71.4^{*}	2.5	40.0	7.5		
Tied migrant	25.0	9.4^{*}	21.4	20.9^*		
Moved for both jobs	50.0	5.2	$62.5^{*\dagger}$	9.0		
Total	22.0	3.0	18.0	6.1		

 Table 4: Employment flows by migrant status: Men and women in couple households

Notes: BHPS 1991-2003. Inflow rates defined as the probability of being in employment at time t, conditional on being out of work at t-1. Outflow rates defined as the probability of being out of work at time t, conditional on being in employment at t-1. * indicates flows significantly different from those for non-migrants at the 5% level. † indicates flows significantly different from that for job-related migrants at the 5% level.

0	l l			
	No earner	Single earner	Dual earner	Total
	couple	couple	couple	
Migrants	8.0	6.9	6.3	6.6
Job migrants	0.4	1.4	1.3	1.3
Moved for husband's job	0.1	1.0	0.6	0.7
Moved for wife's job	0.1	0.2	0.3	0.3
Moved for both jobs	0.2	0.3	0.4	0.3
Ν	1583	6020	17969	25572

Table 5: Migration rates by household employment status

Notes: migration rates. BHPS 1991-2003.

	Spec [1]	Spec	[2]
	Husbands	Wives	Husbands	Wives
Full-time employed t-1	2.358	2.569	2.366	2.589
	[55.73]	[68.10]	[55.69]	[68.25]
Part-time employed t-1	1.785	2.130	1.795	2.146
	[23,10]	[60.50]	[23.20]	[60.82]
Spouse full-time employed t-1	0 345	0 424	0 343	0 421
pouse run time employed t'i	[7 60]	[9 84]	[7 56]	[9 77]
Shouse part time employed t 1	0.204	0.205	0.203	0 203
spouse part-time employed t-1	0.294	[2 94]	0.293	[2 92]
Tab uslated us amount	[0.07]	[3.80]	[0.04]	[3.83]
lob-related migrant	-0.331	-0.414	0.202	0.280
	[2.15]	[3.64]	[0.64]	[1.62]
Employed t-1 and job-related migrant			-0.625	-0.986
			[1.90]	[5.07]
<i>Husband's education</i>				
Degree or above	0.364	0.131	0.362	0.135
	[5.23]	[2.53]	[5.21]	[2.61]
A-Levels or equivalent	0.209	0.120	0.209	0.119
-	[4.59]	[3.08]	[4.58]	[3.05]
GCSEs or equivalent	0.175	0.051	0.174	0.052
1	[3.22]	[1.14]	[3.20]	[1.16]
Other qualifications below GCSEs	0 1 1 0	-0.042	0 109	-0.043
Suler quantientiens below SebEs	[1 57]	[0 72]	[1 56]	[0 73]
Wife more educated than husband	0.030	0.244	0.031	0.742
whe more educated than husband	[0.62]	[6.03]	[0.64]	[5 08]
Cohebiting couple	[0.02]	0.03	[0.04]	[3.96]
Jonabiting couple	0.044	-0.020	0.042	-0.012
.	[0.37]	[0.22]	[0.36]	[0.13]
rivate tenant	0.046	-0.065	0.050	-0.0/4
.	[0.28]	[0.48]	[0.31]	[0.55]
lome owner	-0.154	-0.049	-0.148	-0.055
	[1.25]	[0.49]	[1.21]	[0.55]
lusband's age				
Aged 35-44	0.195	0.103	0.195	0.103
	[4.44]	[3.09]	[4.44]	[3.09]
Aged 45-58	0.060	0.011	0.060	0.010
	[1.45]	[0.29]	[1.43]	[0.29]
Age husband – age wife	-0.018	0.005	-0.018	0.005
6 6	[4.71]	[1.53]	[4.69]	[1.52]
One child	0.217	-0.025	0.218	-0.025
	[2.90]	[0.42]	[2.92]	[0.42]
Fwo children	0.060	-0 041	0.057	-0.036
	[0,70]	[0.62]	[0.67]	[0.56]
Three or more children	[0.70]	0.060	[0.07]	0.069
Three of more children	0.090	0.009	0.101	0.008
(Terrent) 11 Januari 1 - 1 - 5	[0.86]	[0.80]	[0.89]	[0.80]
Has child under 5	0.025	-0.118	0.026	-0.11/
	[0.38]	[2.49]	[0.40]	[2.47]
New child t-1 to t	-0.067	-0.645	-0.065	-0.648
	[0.69]	[9.83]	[0.67]	[9.89]
First child born t-1 to t	0.213	-0.534	0.216	-0.528
	[1.16]	[4.94]	[1.17]	[4.88]
Child left home t-1 to t	-0.081	0.062	-0.081	0.073
	[0.58]	[0.49]	[0.58]	[0.57]
Rho	0.114	0.094	0.113	0.093
	[4.74]	[4.73]	[4.70]	[4.66]
og likelihood	_3973	-6481	_3971	-6468
$2 \sqrt{2} \sqrt{2} \sqrt{2} \sqrt{2} \sqrt{2} \sqrt{2} \sqrt{2} 2$	5115	0.4650	5711	0 6512
p -value $(u_h - u_w)$		0.4050		0.0515
p-value interaction($\alpha_h = \alpha_w$)			4.4.480.80	0.0635
N person years (individuals)		2/2	14 (5050)	

Table 6: Probability of current employment

N person years (individuals) 24214 (5050) Notes: Dynamic random effects probit estimates. Dependent variable takes value 1 if individual is employed at *t* and zero otherwise. Controls also include year dummies, region of residence and means of time-varying covariates (see text for details).

	Spec [1]		Spec [2]
	Husbands	Wives	Husbands	Wives
Lead migrant	-0.056	-0.486	0.880	0.242
-	[0.23]	[1.73]	[1.57]	[0.39]
Tied migrant	-0.750	-0.590	-0.559	0.053
	[2.79]	[4.09]	[0.98]	[0.25]
Moved both jobs	-0.279	0.180	0.365	1.207
	[1.13]	[0.83]	[0.66]	[3.49]
Employed t-1 and lead migrant			-1.117	-0.836
Employed (1 and find mission			[1.8/]	[1.31]
Employed t-1 and tied migrant			-0.231	-0.919
Employed t 1 and moved both jobs			0.786	[3.07] 1 508
Employed t-1 and moved both jobs			-0.780	[3.86]
Rho	0 1 1 3	0.096	0 111	0.094
	[4.70]	[4.79]	[4.65]	[4.72]
Log likelihood	-3969	-6475	-3967	-6460
p-value (joint significance job moves)	0.0310	0.0001	0.2805	0.0064
p-value $(\alpha_{b_1} = \alpha_{b_1})$	0.0561		0.0720	
p-value $(\alpha_{\rm nu} = \alpha_{\rm nu})$		0.7381		0.7702
Cross-equation restrictions				
p-value $(\alpha_{\rm bl} = \alpha_{\rm wt})$		0.0002		0.0001
p-value $(\alpha_{h} = \alpha_{w})$		0.1251		0.3005
p-value $(\alpha_{ht} = \alpha_{wl})$		0.3459		0.1931
p-value $(\alpha_{ht} = \alpha_{wt})$		0.2686		0.0047
Interaction terms:				
p-value (joint significance job moves)			0.1482	0.0000
p-value ($\alpha_{\rm bl} = \alpha_{\rm bt}$)			0.2913	
p-value $(\alpha_{\rm nu} = \alpha_{\rm nu})$				0.9034
Cross-equation restrictions				
p-value $(\alpha_{hl} = \alpha_{wl})$				0.4291
p-value ($\alpha_{hl} = \alpha_{wl}$)				0.6594
p-value ($\alpha_{ht} = \alpha_{wl}$)				0.3413
p-value ($\alpha_{ht} = \alpha_{wt}$)				0.0060
N person years (individuals)		24214	(5050)	

Table 7: Probability of current employment

Notes: see notes to Table 6.

Table 8: Summary of results

	Table 6: Summar	y of results	
	Impact for husband	Impact for wife	Significant difference?
Job-related migrant	* -	*	No
Job-related migrant, not working	+	+	No
Job-related migrant, working	*	* _	Yes
Tied migrant	-*	*	No
Tied migrant, not working	-	+	Yes
Tied migrant, working	-	* _	Yes
Lead migrant	-	*	No
Lead migrant, not working	+	+	No
Lead migrant, working	*	-	No

Notes: Summary of results presented in Tables 6 and 7. ^{*} Indicates statistically significant at the 10% level. Final column indicates whether impact for husband is significantly different from that for the wife.

Husbands	All	Move job	Lead migrant	Tied	Move both
	individuals	reasons		migrant	jobs
Wanted fewer hours	39.9	41.5	45.7	38.3	34.5
Wanted more hours	5.9	8.6	6.2	12.8	10.3
Wanted same hours	54.2	50.0	48.1	48.9	55.2
Weekly work hours	39.4	39.8	41.2	37.6	38.4
Ν	17752	234	129	47	58
Wives					
Wanted fewer hours	32.3	32.3	33.3	30.5	34.6
Wanted more hours	7.1	6.5	8.8	5.7	5.5
Wanted same hours	60.6	61.3	57.9	63.8	60.0
Weekly work hours	28.2	31.8	35.0	30.3	31.4
Ν	17499	217	57	105	55

Table 9: Work hours and work hour preferences at t-1 by migrant status t-1 to t

Notes: BHPS 1992-2003.

¥	Job	Husband's	Wife's	Both
	reasons	job	job	Jobs
Husband wanted to move for job reasons t-1	1.284	1.194	0.080	1.322
	[8.56]	[6.70]	[0.16]	[5.89]
Wife wanted to move for job reasons t-1	1.347	0.950	1.098	1.368
	[7.36]	[3.71]	[3.87]	[5.16]
Both wanted to move for job reasons t-1	2.171	1.425	1.623	2.094
	[8.56]	[4.35]	[4.49]	[6.40]
Husband employed full-time t-1	0.176	0.394	-0.154	0.023
	[1.52]	[2.34]	[0.89]	[0.12]
Husband employed part-time t-1	0.145	0.178	-0.030	0.196
	[0.82]	[0.71]	[0.11]	[0.69]
Wife employed full-time t-1	0.185	-0.070	0.767	0.311
	[2.30]	[0.71]	[3.82]	[2.16]
Wife employed part-time t-1	-0.038	-0.188	0.472	0.043
	[0.46]	[1.94]	[2.29]	[0.29]
Husband's education				
Degree or above	1.006	1.057	0.542	0.985
	[7.88]	[6.34]	[2.65]	[3.44]
A-Levels or equivalent	0.598	0.589	0.283	0.796
	[5.18]	[3.88]	[1.60]	[2.98]
GCSEs or equivalent	0.451	0.336	0.249	0.755
	[3.63]	[2.03]	[1.32]	[2.76]
Other qualifications below GCSEs	0.267	0.109	0.271	0.401
	[1.65]	[0.48]	[1.19]	[1.12]
Wife more educated than husband	0.197	0.187	0.303	-0.023
	[2.29]	[1.61]	[2.28]	[0.15]
Cohabiting couple	-0.015	-0.116	0.084	0.089
	[0.19]	[1.02]	[0.65]	[0.64]
Housing tenure	0.000	0.7(0)	0.505	0 (00
Private tenant	0.809	0.768	0.525	0.689
11	[/.01]	[5.45]	[2.66]	[3.12]
Home owner	-0.190	-0.263	-0.236	0.037
	[1.91]	[2.12]	[1.38]	[0.19]
Husbana's age	0.022	0.010	0.000	0.026
Aged 35-44	-0.032	-0.019	-0.086	-0.036
A 1 45 50	[0.46]	[0.22]	[0.70]	[0.24]
Agea 43-58	-0.181	-0.220	-0.1/5	-0.092
A go husband ago wife	[2.21]	[2.02]	[1.27]	[0.04] 0.025
Age nusband – age whe	0.017	0.010	0.002	[2 21]
Children	[2.31]	[1.93]	[0.21]	[2.21]
One shild	0.012	0.156	0 200	0.002
One child	0.015	0.130	-0.288	-0.003
True shildren	[0.13]	[1.46]	$\begin{bmatrix} 1.74 \end{bmatrix}$	[0.02]
I wo children	-0.008	0.037	-0.123	0.022
Three or more children	$\begin{bmatrix} 0.09 \end{bmatrix}$	0.30	0.769	$\begin{bmatrix} 0.14 \end{bmatrix}$
Three of more children	[1 30]	[1.83]	-0.208	[1 17]
Has child under 5	0.245	0.180	0.268	0.274
Thas chind under 5	[3 23]	[1 02]	[1.81]	[2 01]
Loglikelihood		<u>[1.72]</u> Q1/		.450
Chi ² (joint significance identifying variables)	-1390	70 50	21.62	80 01
$C_{\rm III}$ (joint significance identifying variables) $P_{\rm robs}$ abi ²	102.79	12.30	0.0000	00.01
	0.0000	0.0000		0.0000
in person years (individuals)		24214	+ (3030)	

	Table 10: Probabilit	y of couples moving f	for job related reasons
--	----------------------	-----------------------	-------------------------

Notes: Random effects probit estimates. Dependent variable takes value 1 if couple move for the relevant reason and zero otherwise. Controls also include year dummies and region of residence.

			1	<i>v</i>
	Husbands	Wives	Husbands	Wives
Job-related migrant	-0.600	-0.074	-0.034	0.488
	[1.30]	[0.20]	[0.06]	[1.38]
Job-related migrant and employed t-1			-0.619	-0.977
			[1.88]	[5.02]
Mills ratio	-0.110	0.136	-0.094	0.086
	[0.62]	[0.99]	[0.54]	[0.67]
Log-likelihood	-3973	-6480	-3971	-6468
Lead migrant	-0.622	-0.869	0.642	0.320
-	[0.64]	[0.71]	[0.56]	[0.23]
Tied migrant	-0.501	0.485	-0.406	0.533
C C	[0.42]	[0.86]	[0.34]	[1.01]
Moved both jobs	-1.097	-0.936	-0.346	0.774
-	[1.10]	[0.94]	[0.31]	[0.79]
Lead migrant and employed t-1			-1.085	-0.925
			[1.79]	[1.42]
Tied migrant and employed t-1			-0.240	-0.870
			[0.40]	[3.33]
Moved both jobs and employed t-1			-0.742	-1.564
			[1.21]	[3.73]
Mills ratio moved husband job	-0.212	0.400	-0.079	0.191
	[0.60]	[1.96]	[0.23]	[0.95]
Mills ratio moved wife job	0.055	-0.236	0.020	-0.108
	[0.14]	[0.61]	[0.05]	[0.28]
Mills ratio moved both jobs	-0.286	-0.372	-0.235	-0.140
-	[0.84]	[1.13]	[0.72]	[0.48]
Log-likelihood	-3969	-6475	-3966	-6462
N couple-years (couples)	24214 (5050)			

Table 11: Selection corrected estimates of the p	probability of curren	t employment
--	-----------------------	--------------

Notes: Selection-corrected estimates. Identifying variables are whether or not the husband wanted to move for job-related reasons at t-1, the wife wanted to move for job-related reasons at t-1, and both the husband and wife wanted to move for job-related reasons at t-1. (see Table 10 for details).

Table 12: Summary of results				
	Impact for	Impact for wife	Significant	
	husband	-	difference?	
Job-related migrant	-	-	No	
Job-related migrant, not working	-	+	No	
Job-related migrant, working	*	-	Yes	
Tied migrant	-	+	Yes	
Tied migrant, not working	-	+	Yes	
Tied migrant, working	-	-	No	
Lead migrant	-	-	No	
Lead migrant, not working	+	+	No	
Lead migrant, working	-*	-	No	

Notes: Summary of results presented in Table 11. ^{*} Indicates statistically significant at the 10% level. Final column indicates whether impact for husband is significantly different from that for the wife.